Author’s response to reviews

Title: Bioinformatics analyses on the immune status of renal transplant patients, a systemic research of renal transplantation

Authors:

Mei Meng (17211210066@fudan.edu.cn)
Weitao Zhang (18111210072@fudan.edu.cn)
Qunye Tang (tang.qunye@zs-hospital.sh.cn)
Baixue Yu (18211210067@fudan.edu.cn)
Tingting Li (19211210070@fudan.edu.cn)
Ruiming Rong (rong.ruiming@zs-hospital.sh.cn)
Tongyu Zhu (tyzhu@fudan.edu.cn)
Ming Xu (xu.ming@zs-hospital.sh.cn)
Yi Shi (shi.yi@zs-hospital.sh.cn)

Version: 2 Date: 20 Nov 2019

Author’s response to reviews:

1. Authorship Change Form

We note that the order of authors has changed since original submission of the manuscript. In line with COPE guidelines, BioMed Central requires written confirmation from all authors that they agree with any proposed changes in authorship of submitted manuscripts or published articles.

In such cases, we use a standardised form which we would request you and your co-authors to complete. The authorship change form can be found at the following link: https://resource-cms.springernature.com/springer-cms/rest/v1/content/7454878/data/v5

All instructions can be found on the form, please treat the 'current authorship' section as the original authorship. Please return the form within 14 days by email to the Editorial Office with all author signatures (including those newly added/removed).
We sent the form on Oct 8th. To make sure it arrives, we sent it again this morning.

Miss Li Tingting has been a Mphil student in my lab since Sept 1st. She helped us go through all the data sets and relevant literature. All authors appreciated her contribution and agreed to add her as a co-author in the study. Miss Li is listed as 6th authors

2. Availability of data and materials

Your current statement in the Availability of data and materials section is incorrect. Although you mentioned that you downloaded genomes from the NCBI Geo database in the Methods section of your manuscript, you have not provided the relevant accession numbers. Please provide the relevant accession numbers or the appropriate web links to the all of the data sets used in your study in the Availability of Data and Materials section of your manuscript.

Your data availability statements can take the following form: “The datasets generated and/or analysed during the current study are available in the [NAME] repository, [PERSISTENT WEB LINK TO DATASETS].”

We have added the detail information in supplements (S table 1) and added the statement in the manuscript

3. Funding

a) Please provide the full name of the funding body corresponding to the grant number “ZSYXRC-014”.

Zhongshan Hospital Seed Funding

b) Please provide more details regarding the specific role of each of the funding bodies in your study. If the funding bodies only provided the financial means to allow the authors to carry out the study, please state “The funding bodies played no role in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript” and include this statement in the Funding section of your manuscript.
However, if each of the funding bodies took part in the design of the study and collection, analysis, and interpretation of data, and the writing of the manuscript, this must be stated in the Competing Interests section of your manuscript.

We have added the statement in the manuscript

4. Additional files

As per our submission guidelines, we also require that all additional files be individually cited in the text in sequence in a manuscript. Your supplementary tables are currently not cited in your manuscript. Please ensure that they are appropriately cited within the main body of text in your manuscript. Any items which are not cited may be deleted by our production department upon publication.

We have cited the supplementary table accordingly

5. Declarations heading

Please ensure that you include a ‘Declarations’ heading and that it is placed after the Abbreviations in your manuscript.

The Declarations section encompasses the sub-sections: Ethics approval and consent to participate, Consent for publication, Availability of data and material, Competing interests, Funding, Authors’ contributions, Acknowledgements.

We have added the head 'Declaration'

6. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Should you wish to respond to these revision requests, please put your responses to the reviewers'/editors’ comments in the Response to Reviewers box in Editorial Manager. Please do not upload a separate letter.

The clean manuscript has been uploaded